



Tourism Destination Performance and Competitiveness: The Impact on Revenues, Jobs, the Economy, and Growth

Rudi Purwono Department of Economics, Airlangga University , Surabaya, Indonesia Miguel Angel Esquivias Department of Economics, Airlangga University , Surabaya, Indonesia Lilik Sugiharti Department of Economics, Airlangga University , Surabaya, Indonesia Omar Rojas Facultad de Ciencias Económicas y Empresariales, Universidad Panamericana, Zapopan, Mexico

Received: 24 August 2023. Revision received: 11 November 2023. Accepted: 15 January 2024

Abstract

We analyzed the relationship between tourism destination competitiveness (TDC) and economic performance using the Travel and Tourism Development Index (TTDI) and various tourism metrics. Employing a Structural Equation Model (SEM) and the latest available 2021 data, we focused on factors such as connectivity, infrastructure, demand drivers, sustainability, and the enabling environment. The findings indicated a positive correlation between these factors and the pillars of destination competitiveness. Key elements such as connectivity, infrastructure, demand drivers, sustainability, and enabling environment significantly influence TDC in Asia. Our research demonstrates that these indicators strongly affect various aspects of performance in Asian countries, including tourism arrivals, revenue, and job creation. We found a negative correlation between certain aspects of destination performance and tourism growth, suggesting that highly competitive destinations may experience lower growth rates than less competitive ones, thereby affecting the overall development of the sector. This highlights the necessity of specific strategies to leverage the positive influence of competitive tourist destinations on economic dynamics and job creation within a destination. This study highlights the crucial role of core resources and infrastructure in enhancing tourism performance and competitiveness, and advocates targeted efforts for sustainable and resilient sector growth. These findings offer valuable insights into the relationship between competitiveness and performance in the Asian tourism industry.

Key Words: Tourism competitiveness; job creation; tourism market development; sustainable tourism; tourism employment; infrastructure

JEL Classification: O11; L83; O14

Reference: Purwono, R., Esquivias, M. A., Sugiharti, L., & Rojas, O. (2024). Tourism Destination Performance and Competitiveness: The Impact on Revenues, Jobs, the Economy, and Growth. *Journal of Tourism and Services*, 15(28), 161–187. https://doi.org/10.29036/jots.v15i28.629

1. Introduction

In 2019, international tourism arrivals reached nearly 1.5 billion trips, as reported by the World Tourism Organization (UNWTO), marking a 4% increase over 2018 and more than doubling the number recorded in 2000 (less than 700 million). In Asia, tourism contributes to almost 10% of the GDP, creating over 160 million jobs. International tourism receipts in Asia reached nearly 480 billion dollars in 2019. Countries like Bangladesh, Indonesia, the Philippines, and Malaysia nearly tripled their

Scopus



tourism receipts between 2000 and 2019. India, Sri Lanka, Thailand, and Vietnam saw their international receipts rise sevenfold in the past two decades.

The increasing importance of tourism as a revenue generator, job creator, source of foreign currency, and magnet for investment has attracted both academic and policymaking attention. Scholars have explored factors influencing tourism's increasing contribution to economic growth (Michael et al., 2019; Gavurova et al., 2021), a phenomenon often investigated under the tourism-led growth hypothesis (Alkathiri & Soliman, 2020; Adeleye, 2023). Considerable focus has been on evaluating tourism competitiveness as a factor that can contribute to destination attractiveness (Kim et al., 2022; Mior Shariffuddin et al., 2022; Zadeh Bazargani & Kiliç, 2021), as well as factors enabling it (Stankova et al. 2019; Kurar, 2021). Competitiveness also increases the quality of offerings (Civelek et al., 2023a) and adoption of digital technologies (Civelek et al., 2023b). Although some studies have developed comprehensive frameworks that include indicators of tourism competitiveness, there is limited research examining the relationship between tourism competitiveness and performance (e.g., Hanafiah & Zulkifly, 2019; Zadeh Bazargani & Kiliç, 2021).

Shifts in tourism competitiveness may not consistently align with the economic dimension of destination performance, encompassing aspects such as visitor counts, market share, tourist spending, employment, and value added by the tourism industry (Abreu Novais et al., 2018; Heath, 2003). The literature highlights instances where improvements in destination competitiveness do not result in an increase in the number of visits, earnings from foreigners, employment opportunities, or investments in these locations. For instance, the Mediterranean region (Kunst & Ivandić, 2021) highlights this. Similarly, Croes and Kubickova (2013) noted that achieving a high ranking in the competitiveness index, based on indicators from the World Economic Forum (WEF), does not correlate with higher receipts or improved quality of life for residents in the destination country. Webster and Ivanov (2014) found that destination competitiveness did not significantly impact economic growth. Furthermore, Dwyer (2022b) emphasized that many tourism competitiveness models frequently overlook the link between destination competitiveness and residents' well-being. Although the correlation between destination competitiveness and the economic dimension of tourism competitiveness may not be immediately apparent, it can still affect factors such as "attractiveness and satisfaction" with the destination (Dwyer & Kim, 2003; Syofian et al. 2023) or sustainability (Mior Shariffuddin et al., 2022).

As not all destination competitiveness attributes equally promote high-performing tourism destinations, it is essential to evaluate which factors are more critical. Policymakers often rely on competitiveness indicators to evaluate the success of their destinations as tourist attractions (Domínguez Vila et al., 2015; Drakulić Kovačević et al., 2018; Lesmana et al., 2022). These indicators capture concepts, often in the form of indexes, but they do not suggest a "cause-effect relationship" or a "link between inputs and outputs." For example, appealing core resources (natural attractions, cultural sites, or unique species) or supporting conditions in place (ATMs, hotel rooms, or a visa-on-arrival) can be seen as possessing competitive attributes conducive to thriving tourism, according to the Travel & Tourism Development Index (ITDI) indicators. Nonetheless, assessing tourism performance metrics from an economic perspective (visitor arrivals, earnings, job creation, or contribution to GDP) should be an integral part of defining destination competitiveness and provide a more robust benchmark for evaluating competitiveness. Thus, it is crucial to investigate the relationship between attractiveness and tourism performance.

As tourism competitiveness assessment relies on proxy indicators, employing factor analysis with a reflective-measured property approach is a potential method for scrutinizing the link between competitiveness and performance. We further investigated this aspect by utilizing a structural equation model using the Partial Least Square - Structural Equation Model PLS-SEM approach. Supported by the literature on tourism destination competitiveness (TDC), we assessed whether TDC attributes could help explain tourism performance in Asian countries. We compared two sets of destination competitiveness indicators commonly employed by policymakers to determine the tourism





attractiveness of their countries: the TTDI and alternative destination competitiveness (TDI) indicators (including indicators from the UNWTO and others). Moreover, we employed a set of indicators to measure tourism performance, considering aspects related to tourism growth, contributions to destinations, arrivals, receipts, and financial performance indicators (i.e., income, job creation, exports, investment, and business expenditure).

This study makes several contributions. First, we employed two distinct models with different datasets to investigate the relationships between various destination factors and tourism competitiveness. The results highlight the positive contributions of several factors to destination performance—connectivity, infrastructure, travel and tourism demand drivers, sustainability, core resources, enabling environment, and the pillars of tourism competitiveness. Second, by delving into the relationship between TDC measures and destination performance, our analysis helps to understand whether increased competitiveness aligns with improved economic and tourism performance. Third, the findings underscore the pivotal role played by core resources and infrastructure in driving high destination performance. These elements are instrumental in increasing tourist numbers and revenue, and creating direct employment opportunities. Fourth, it emphasizes the substantial contribution of key destination aspects to the tourism sector's financial impact, incorporating expenditure, exports, business activity, and GDP growth, showcasing the broad economic impact of a well-developed tourism strategy. Our primary contribution pertains to evaluating whether certain tourism destination indicators utilized by policymakers effectively serve as benchmarks for guiding tourism competitiveness.

The remainder of this paper is structured as follows: Section 2 presents the theoretical background; Section 3 describes the Materials and Methods employed; Section 4 presents the results; and Section 5 presents the conclusions, including limitations and possible scope for future research.

2. Theoretical background

Four foundational theoretical studies have paved the way for the literature on TDC as an effective model for assessing tourism competitiveness (Dwyer & Kim, 2003; Gooroochurn & Sugiyarto, 2005; Heath, 2003). Subsequent research has expanded the conceptual and empirical understanding. The Crouch model (Crouch, 2011) emerged as a key reference, encompassing factors such as "core resources and attractions, supporting elements, qualifying and amplifying determinants, destination management, and destination policy." Numerous efforts have been devoted to devising indicators and establishing frameworks for evaluating destination competitiveness (Croes, 2011; Croes & Kubickova, 2013; Leung & Baloglu, 2013; Krivokapic, 2023). Disputes exist regarding concepts (Dwyer, 2022a; Luštický & Štumpf, 2021), indicators (Gooroochurn & Sugiyarto, 2005; Nadalipour et al., 2019), measurements (Aguiar-Barbosa et al., 2021; Pulido-Fernández & Rodríguez-Díaz, 2016; Perez Leon et al., 2021), and models (Mior Shariffuddin et al., 2022; Nematpour et al., 2022).

While the prevailing belief centers on TDC factors from destinations' comparative and competitive advantages (Domínguez Vila et al., 2015; Dwyer & Kim, 2003), several reasons underscore the necessity for new comparative studies. First, the dynamism inherent in tourism competitiveness calls for more recent indicators, concepts, and models to yield more precise measurements. Second, aspects related to escalating globalization, shifts in tourism pricing, national economic development trajectories, and the extent to which countries prioritize tourism earnings (Croes and Rivera, 2010; Y. R. Kim et al., 2022) often elude the literature's scope. Third, prevailing theoretical models on tourism demand overlook elements such as globalization, openness, and foreign direct investment as dimensions of competitiveness (Hanafiah & Zulkifly, 2019; N. Kim & Wicks, 2010). Furthermore, tourism performance is frequently limited to metrics linked to tourism arrivals and income within specific periods (Zadeh Bazargani & Kiliç, 2021), neglecting dynamic facets (rate of growth, market





share fluctuations, alterations in foreign income contribution, among others) and omitting the evaluation of impacts on economic variables like job generation, investment, exports, or societal contributions (Aguiar-Barbosa et al., 2021; Dwyer, 2022a; Luštický & Štumpf, 2021; Gavurova et al., 2021). Measuring performance solely through arrivals or income generation fails to capture tourism's holistic contribution to the economy (i.e., jobs, exports, etc.), or assess the destination's long-term sustainability (both in terms of growth and significance) (Hanafiah & Zulkifly, 2019).

Our first set of destination competitiveness indicators were published by the TTDI, while the second was developed from the TDC indicators proposed by Hanafiah and Zulkifly (2019). Regarding tourism performance indicators, we encompassed five categories to summarize the upsurge in tourism activities: economic contributions (GDP, exports, employment, and consumption), tourism destination achievements (arrivals and receipts), and financial performance (expenditure, capital investment, and direct and indirect GDP contributions). We employed data from 35 Asian countries to offer a robust benchmark for Asia. Our analysis incorporated the TTDI 2021 dataset, and an additional set of competitiveness indicators as suggested in previous studies (Drakulić Kovačević et al., 2018; Hanafiah & Zulkifly, 2019; Y. R. Kim et al., 2022).

The ability to grow at a faster pace and exhibit much better performance depends on destination competitiveness. Tourism performance indicators include tourism satisfaction (Grissemann & Stokburger-Sauer, 2012; Syofian et al., 2023; Ginting et al., 2023), arrivals (Heriqbaldi et al., 2023), income (Zadeh Bazargani & Kiliç, 2021), value added (Ragab & Meis, 2016), well-being (Dwyer, 2022a), and quality of life (Croes & Kubickova, 2013). Economies can increase income by improving community competitiveness through a consideration of internal and external factors (Hanafiah & Zulkifly, 2019).

A tourist destination's competitiveness is influenced by different internal and external factors that determine its general performance (Rodríguez-Díaz & Pulido-Fernández, 2019). Furthermore, as the competitiveness of destinations increases, the impact on economic growth and inhabitants' prosperity is expected to increase, consistent with the conceptual definition of competitiveness of Croes and Kubickova (2013) and Ritchie and Crouch (2003). Although economic performance is often considered in the causal chain of competitiveness (Croes, 2011; Gavurova et al., 2021), testing whether competitiveness is related to performance seems more appropriate. Furthermore, a comprehensive assessment of destination competitiveness rooted in performance metrics provides a more direct and responsive framework for shaping tourism policies, refining evaluation processes, enhancing promotional efforts, facilitating learning, and improving decision-making.

Previous studies have utilized the TTDI or its equivalent indicators to evaluate destination competitiveness. For example, Lesmana et al. (2022) built a framework to test indicators from the TTCI and UNWTO "Tourism Dashboard" to assess Indonesia. The authors identified 63 indicators within 12 dimensions with potential links to Indonesia's competitiveness as a tourism destination. However, the framework failed to test whether such competitiveness factors were related to performance (e.g., arrivals, receipts, jobs, and exports). Vila et al. (2015) tested the Crouch model in Australia and Spain, revealing significant disparities in the factors and attributes that impact competitiveness within each destination. Some studies linking competitiveness and performance include Zadeh Bazargani & Kiliç (2021) in which performance was proxied by receipts and arrivals. By employing the TTDI, indicators alongside destination performance metrics (arrivals, spending, T&T share of GDP, and employment), Rodríguez-Díaz and Pulido-Fernández (2019) found a positive correlation between tourism performance and competitiveness indicators, particularly pronounced within the resources category, whereas a comparatively weaker connection was observed within the T&T policy and conditions domains. Leung and Baloglu (2013) utilized the TTCI indicators from 2011 to assess the destination competitiveness of Asian countries, presenting clusters of competitiveness. While they underscored the diverse strengths of Asian countries, with significant practical implications, they did not explore the correlation between these clusters and destination economic performance.





Understanding the intricate relationship between competitiveness and performance is crucial to develop effective strategies and policies for international community development through tourism. However, the existing literature has predominantly examined instruments, models, and indicators or offered insights specific to particular locations, leaving a gap in comprehending the interrelationship between destination competitiveness and performance in an Asian context.

3. Materials and Methods

This study investigated two factors: the validity of tourism destination indicators commonly utilized by policymakers as proxies for guiding tourism competitiveness, and the impact of tourism competitiveness indicators on tourism performance. We presented two models to compare the competitiveness of different tourist destinations. We began by presenting the constructs used to assess tourism performance. Subsequently, we elaborated on two alternative frameworks of competitiveness to examine the correlation between competitiveness and tourism performance within Asia. To gauge tourism performance, we identified three constructs (dimensions) of performance variables that encompass aspects of tourism growth, economic contribution (Hanafiah & Zulkifly, 2019), and financial performance (Nematpour et al., 2022).

Table 1 provides an overview of these constructs along with their respective indicators and definitions. The tourism growth construct (referred to as growth) encompasses indicators presented as percentages that capture internal travel and tourism consumption growth, business tourism spending, government expenditure on tourism services, and growth in foreign tourism spending. The second, centered on tourism destination financial performance, includes tourism arrivals, tourism receipts, and employment creation indicators.

Construct	Indicator (Definition)	Variable name
	Business Tourism Spending % growth	BTS1
Tourism	Tourism Spending % growth	dts1
(Growth)	Government spending on T&T service % growth	GST1
	T&T's direct contribution to GDP % growth	ttdcgdp1
Financial Performance (Value \$)	International tourism, number of arrivals million	Arrivals
	International tourism, receipts (% of total exports)	TTCR5
	T&T's direct contribution to employment Thousands of jobs	Ttdce3
Economic contribution (Share GDP)	Government spending on T&T service % total tourism expenditure	GSTT2
	T&T's direct contribution to GDP % share of total GDP	TTDCGDP2
	Visitor Exports (Foreign spending) % exports	VE1

Table 1. Constructs for Tourism Performance Indicators

Source: WTTC and EIR raw data, IMF (IMF, 2021), UNDP (UNDP, 2021).

The third assesses economic performance by incorporating indicators regarding government investment, direct and total contributions to GDP, and various facets of economic contribution, including job creation (both direct and total), the contribution of tourism to GDP, and contribution of travel and tourism (T&T) to total spending. While most indicators were sourced from reputable entities such as the World Travel & Tourism Council (WTTC) through the Economic Impact Reports (EIR), International Monetary Fund (*IMF*, 2021) report, and United Nations Development Programme (*UNDP*, 2021) annual report, additional performance indicators were explored. These included metrics associated with capital investment, business expenditure, travel, and tourism contributions to total





employment, and exports. However, some additional indicators encountered challenges in terms of reliability and consistency.

Tourism competitiveness was assessed using two distinct frameworks. The first utilized the TTDI 2021 as an indicator of tourism competitiveness, encompassing five constructs composed of 29 indicators. We initially conducted tests to ascertain the presence of a significant correlation between these indicators and destination competitiveness. After verifying the validity of the framework and the suitability of the indicators for reliable measurements, we evaluated whether competitiveness-related constructs exhibited a significant relationship with tourism performance. Table 2 provides a comprehensive overview of the indicators used for the seven constructs of competitiveness derived from the TTDI 2021 and performance constructs.

Construct	Indicator	Variable name
	Health and Hygiene	Health Env
Enabling	Safety and Security	Safety Env
Environment	Business Environment	business
(ENABLING)	ICT Readiness	ICT
	Human Resources and Labor Market	labor
	Tourist Service	Infra 1
Infrastructure (INF)	Ground and Port	Infra 2
	Air Transport	Infra 3
Travel and Tourism	Natural Resources	TTDemand1
Demand Drivers	Cultural Resources	TTDemand3
(DEM)	Non-Leisure Resources	TTDemand
Traval and Tourism	Socioeconomic Resilience & Conditions	Socioeconomic
Sustainability (SUS)	T&T Demand Pressure & Impact	Env Management
oustainability (000)	Environmental Sustainability	ENV Sustainability
	Air connectivity score	Air Transp
	Number of regional trade	FTAs
CONINI	Financial Openness	Openness
(COININ)	Ground Transport	Ground Tra
	Logistic Index	Logistics
	T&T Development Index demand	T_demand
D:11	T&T Policy & Enabling Conditions	TTDI2
Pillars	T&T Demand Drivers	ttdi4
	Enabling Environment	TTDI6

Table 2. Constructs for Tourism Indicators (Pillars based on TTDI 2021)

Note: Travel and Tourism (T&T) Source: own elaboration

Variations in the competitiveness models stem primarily from the incorporated indicators. The TTDI model, structured around tourism pillars, presents two distinct constructs: one centered on the enabling environment for tourism and the other on indicators addressing tourism policy. In contrast, the TTCI offers a range of indicators concerning complementary conditions and destination management, a more comprehensive array of sustainability-focused indicators, and a broader scope encompassing infrastructure-connectivity indicators. Nonetheless, the distinctive sets of indicators within the two alternative models have demonstrated relatively modest explanatory efficacy in determining destination competitiveness (Gómez-Vega & J Picazo-Tadeo, 2019; Michael et al., 2019;





Rodríguez-Díaz & Pulido-Fernández, 2019; González-Rodríguez et al., 2023; Mwinuka & Mwangoka 2023). Our objective was to assess the potency of these indicators within this framework.

In the second framework, we employed eight constructs that cover the framework proposed by Hanafiah and Zulkifly (2019) and gathered indicators from different sources (See Table 3). The eight constructs were 1) Complementary Conditions, 2) Connectivity, 3) Core resources and attractions employed in previous studies (González-Rodríguez et al., 2023; Moradi et al., 2022; Nobre & Sousa, 2022), 4) Destination Management (Michael et al., 2019; Sul et al., 2020), 5) Infrastructure (Azzopardi & Nash, 2016; Sul et al., 2020; Mwinuka & Mwangoka 2023), 6) Sustainability, 7) Tourism pillars borrowed from the TTDI, and 8) Openness. Table 3 presents the indicators; however, not all met the required consistency and reliability standards for our use.

Construct	Indicator	Variable Name
	Hotel room	cc1
	ATM	cc2
	Sanitation	cc3
Complementary Conditions	Internet	cc5
	Broadband internet	cc 7
	Mobile broadband	cc8
	Air connectivity score	CONNECT1
Connectivity	Number of regional trade	FTAs
	Logistic index	Logistic Index
	Natural sites	cr1
Core Resources	Cultural sites	cr2
Core Resources	Species	cr3
	Association meeting	cr5
	Tourist Service Infrastructure pillar	service
	Quality education	dm4
Destination Management	Specialized training	dm6
	Environmental management	dm7
	Air transport	inf1
Infrastructure	ІСТ	ICT
	Ground transport	inf4
	Financial openness	Financial openness
Openness (Global)	FDI Inflows (% of GDP)	FDI_GDP
	Trade (% of GDP)	Trade_GDP
	T&T Demand Drivers	TT_Demand
Dillara	Enabling Environment subindex	ENV
Fillars	Infrastructure subindex	INF
	Sustainability	SUST
	International Openness pillar	Openness
Policy	T&T Policy	Policy
	Prioritization of the T&T pillar	Priority

Table 3. Constructs and Indicators TTCI 2021

Note: Travel and Tourism (T&T) Source: own elaboration





3.1 Conceptual Framework: Hypotheses

To discern the influence of various constructs on tourism competitiveness and their subsequent impact on tourism performance, we formulated two sets of hypotheses (one for each model) to address the research problems encountered in this study. Both models were constructed based on previous research that examined the connections between competitiveness and factors including connectivity, infrastructure (Michael et al., 2019; Fernández et al., 2020), conducive tourism environments, demand considerations, sustainability, core resources (Michael et al., 2019), destination management (Crouch & Ritchie, 1999), and openness. Similarly, the relationship between tourism competitiveness and destination performance has been explored in various studies. Our hypotheses were formulated based on four fundamental theoretical studies that have pioneered the literature on TDC in tourism competitiveness measurement (Dwyer & Kim, 2003, 2003; Gooroochurn & Sugiyarto, 2005; Heath, 2003), and others (Dwyer, 2022a; Grissemann & Stokburger-Sauer, 2012; Zadeh Bazargani & Kiliç, 2021; Ragab & Meis, 2016; Croes & Kubickova, 2013).

Figure 1. Tourism competitiveness model employing TTDI 2021



First model

H1: Connectivity factors positively and significantly influence Asia's infrastructure competitiveness.

H2: Infrastructure factors positively and significantly influence Asia's tourism pillars of competitiveness in enhancing tourism performance.

H3: Enabling tourism environment factors positively and significantly influence Asia's tourism pillars of competitiveness to enhance Asia's tourism performance.

H4: T&T Demand factors positively and significantly influence Asia's pillars of tourism competitiveness in enhancing tourism performance.

H5: T&T Sustainability factors positively and significantly influence Asia's tourism pillars of competitiveness, enhancing tourism performance.





H6: Tourism pillars of competitiveness positively and significantly influence Asia's tourism performance regarding Growth, Share, and economic value.

H6 uses three distinct constructs: a) Tourism Growth (Growth), b) Financial Performance (Value \$), and c) Economic Contribution (Share GDP).

Second model

H7: Connectivity factors positively and significantly influence Asia's infrastructure competitiveness

H8: Complementary conditions positively and significantly influence destination management competitiveness in Asia.

H8: Infrastructure factors positively and significantly influence Asia's tourism pillars of competitiveness, enhancing its tourism performance.

H10: Destination Management factors positively and significantly influence Asia's tourism pillars of competitiveness in enhancing tourism performance.

H11: Core Resource factors positively and significantly influence Asia's pillars of tourism competitiveness in enhancing tourism performance.

H12: Openness factors positively and significantly influence Asian tourism performance (Share GDP).

H13: Economic contributions (Share GDP) positively and significantly influence Asia's tourism policies.

H14: Tourism pillars of competitiveness positively and significantly influence Asia's tourism performance regarding Growth, Share, and economic value.

H14 uses three distinct constructs: a) Tourism Growth (Growth), b) Financial Performance (Value \$), and c) Economic Contribution (Share GDP).

Figure 2. Tourism competitiveness model employing T&T Competitiveness Report (TTCR 2021)



Source: own elaboration

3.2 Empirical analysis





We employed a PLS-SEM approach to examine the proposed empirical framework. PLS-SEM, as advocated by Hair et al. (2012), integrates principal component analysis and path analysis techniques seamlessly (Shariffuddin et al., 2023). Utilizing a principal component analysis enables the development of a reliable system for examining linear correlations between hidden constructs and observed variables (Chin et al., 2003). Considering our empirical strategy's predictive focus, the modest sample size (Wong, 2013), and potential non-normality in data distribution (Wong, 2013), PLS-SEM was a suitable option model evaluation. In addition, PLS-SEM is advantageous for conducting causal predictive analyses in complex situations with limited theoretical knowledge (Bamgbade et al., 2018; Hanafiah & Zulkifly, 2019). The inherent soft approach of PLS allows it to effectively account for theoretical and measurement conditions, distributional considerations, and practical factors (Barclay et al., 1995). Following the procedure recommended by Henseler et al. (2009), we evaluated the outer measurement model and inner structural model in PLS using a two-step process.

A concise description of the research steps can be summarized in the following steps:

- Step 1: Conduct an extensive literature review of tourism competitiveness and identify key theoretical approaches, empirical frameworks, measurements, indices, and data sources.
- Step 2: Formulate hypotheses aligned with the research objectives and propose two distinct frameworks with similar constructs but different indicators and data.
- Step 3: Collect data from various sources to identify indicators representing the constructs outlined in the hypotheses.
- Step 4: Evaluate various software options for conducting the required tests based on the empirical framework and the nature of the chosen constructs and indicators.
- Step 5: Rigorously assess the validity and reliability of the chosen framework (constructs and indicators).
- Step 6: Test hypotheses using the structural equation approach.

4. Results

We first conducted essential reliability, internal consistency, convergent validity, and discriminant validity tests to ensure the integrity of the measurement model components (Hair Jr et al., 2021), following Cenfetelli and Bassellier (2009) and Wong (2013). To ensure reliability, we examined the outer loading factors for each indicator and found that all met the minimum requirement of 0.5, as reported by Wei et al. (2021). In terms of internal consistency reliability, we evaluated the rho alpha values, composite reliability (CR), and Cronbach's alpha values (Hair Jr et al., 2021) to ensure accuracy. The latent construct values exceeded the minimum threshold of 0.7 and the reliability of the scales was reasonably high with Cronbach's alpha and CR greater than 0.7 (Nematpour et al., 2022). The convergence validity was confirmed by calculating the average variance extracted (AVE) for the various constructs proposed, following Fornell and Larcker (1981). It has been suggested that the AVE value should be greater than 0.5, meaning that the latent structure construct should account for at least 50% of the variation observed in the variables (Hair et al., 2012). The results for convergent validity, reliability, and loading factors (internal consistency) for tourism performance constructs and tourism competitiveness models 1 and 2, are presented in Tables 4, 7, and 8. Only the indicators that fulfilled the reliability and validity criteria are presented.

We show the results for each model in the following order: we presented and evaluated the validity of the inner structural model, then tested some main statistics for the external model to prove the 14 hypotheses, six for Model 1, and the rest for Model 2. Once the results were obtained, we discussed them and concluded with recommendations for policymaking.

4.1. Model 1: Tourism Indicators (Five Pillars) – inner structural model





Model 1 assessed the impact of the five constructs of competitive tourism pillars: T&T Policy and enabling conditions, infrastructure, connectivity, T&T demand drivers, and T&T sustainability (Figure 1). The strength of the relationships between the indicators and their corresponding latent constructs was assessed by examining the loading factors. These verified whether the indicators effectively contributed to the measurement of the latent construct (Sarstedt et al., 2014). Loading factors above 0.50 or 0.70 (Hair et al., 2012) are commonly used as a minimum threshold to ensure the indicators' meaningful association with the underlying variables. Our loading factors fulfilled the minimum requirement of meaningful indicators.

We used both Cronbach's alpha values and CR to examine internal consistency and construct reliability. The results of both methods yielded similar outcomes, with all values exceeding the threshold of 0.7. This indicates that all constructs met the required parameters, affirming their reliability. To assess convergent validity, we examined the AVE values for the proposed constructs, adopting a minimum threshold of 0.5, as recommended (Hair et al., 2012). AVE values of at least 0.5 indicate that the constructs account for more than 50% of the variance in the indicators (Table 4). The obtained AVE values greater than 0.6 suggest that the model fulfilled the requirements for convergent validity and demonstrated good internal consistency. This study assessed discriminant validity using the Fornell-Larcker criteria (1981). According to this criterion, a construct's square root of AVE should surpass its correlation values with other constructs.

4.2 Evaluation of the inner structural models

Once the minimum criteria for reliability and consistency were met and the indicators and latent constructs effectively represented the framework, we conducted a path analysis to determine whether the five aspects collectively shape a competitive destination and lead to significant tourism performance. To evaluate the predictive relevance of the models, we employed various metrics, including Q2, goodness-of-fit index (GOF), coefficient of determination (R2), path coefficient (B-value), and t-statistics, as suggested by Nematpour et al. (2022). To assess the quality of the path model and cross-validity redundancy, we followed Tenenhaus et al. (2005). The Q2 values for the constructs were greater than zero, indicating the predictive relevance of the model.

The model fit measures obtained through GOF were used specifically to assess whether the model adequately fit the empirical data and whether it was "parsimonious and plausible" (Henseler et al., 2016). GOF values range from 0 to 1, with values below 0.1 considered small fitness, around 0.25 as medium, and above 0.36 as large (Nematpour et al., 2022). The GOF index for the first model was calculated to be 0.573, indicating good predictive power (with AVEs of 0.68 and average R2 values of 0.483) (Tenenhaus et al., 2005). The Small Sample Root Mean Square Residual (SMRS) also fit well. The Normed Fit Index (NFI) was slightly below 0.9 but still fell within an acceptable range. Lohmöller (2013) suggested that NFI values are acceptable at or above 0.9. NFI values closer to one were more favorable. To assess this discrepancy, d_ULS and d_G values were compared with their respective confidence intervals (Henseler et al., 2016). A good fit is obtained if the confidence interval of the upper bound is greater than the original values of d_ULS and d_G.

Table 4.	Validity and	reliability test	t results fron	n tourism c	competitiveness	indicators
	2	2				

	Loading Factor	Cronbach's Alpha	rho_A	Composite Reliability (CR)	Average Variance Extracted (AVE)
Connectivity		0.894	0.906	0.922	0.703
Air Transp	0,887				
FTAs	0,743				



JOURNAL OF TOURISM AND SERVICES

Issue 28, volume 15, ISSN 1804-5650 (Online) www.jots.cz



Ground Transp	0,867				
Logistics	0,879				
Openness	0,809				
T&T Demand		0.872	0.991	0.916	0.787
TTDemand	0,957				
TTDemand1	0,735				
TTDemand3	0,950				
Enabling Enviro	nment	0.910	0.931	0.933	0.738
Business	0,888				
Health Env	0,770				
ICT	0,907				
Labor	0,936				
Safety Env	0,782				
Infrastructure		0.769	0.794	0.867	0.687
Infra 1	0,909				
Infra 2	0,846				
Infra 3	0,719				
Pillars of Competi	tiveness	0.774	0.816	0.858	0.608
TTDI2	0,812				
TTDI6	0,683				
t_demand	0,627				
ttdi4	0,956				
T&T Sustainability		0.720	0.800	0.841	0.643
ENV Sustainability	0,621				
Env Management	0,898				
Socioeconomic	0,859				
Tourism Growth		0.788	0.857	0.863	0.615
BTS1	0,831				
GSTT1	0,672				
TTDCGDP1	0,698				
dts1	0,911				
Share GDP		0.792	0.943	0.871	0.693
GSTT2	0,754				
TTTCEMPL2	0,823				
VE1	0,912				
Value \$		0.767	0.790	0.895	0.810
Arrivals	0,878				
TTCR5	0,921				
Ttdce3	0,886				

Source: own elaboration

Table 5 presents the results of the discriminant validity tests, as per the procedure outlined by Fornell and Larcker (1981). Validation of the measurement model's latent construct correlation confirmed the acceptability of discriminant validity.

Table 5. Fornell-Larcker criteria for tourism competitiveness model employing TTDI 2021

CONN DEM ENABLI Growth INF Pillars SUS Share V:	Value
---	-------





			NG					GDP	\$
CONN	0.847								
DEM	0.673	0.827							
ENABLIN	0.724	0.253	0.823						
G									
Growth	0.241	0.185	0.440	0.784					
INF	0.806	0.676	0.780	0.383	0.856				
Pillars	0.819	0.707	0.807	0.408	0.819	0.822			
SUS	0.673	0.282	0.812	0.508	0.719	0.779	0.802		
Share GDP	0.267	0.376	0.243	0.358	0.341	0.358	0.221	0.832	
Value \$	0.591	0.780	0.247	0.061	0.557	0.565	0.136	0.114	0.841

Source: own elaboration

4.3 Estimation of R2, B, and T-values of the Model

The results indicate that all five aspects of tourism competitiveness (constructs) in the initial results significantly and positively impact the shaping of the tourism sustainability pillars. Table 6 shows that the constructs of Connectivity (CONN), Infrastructure (INF), Travel and Tourism Demand Drivers (DEM), Travel and Tourism Sustainability (SUS), Enabling Environment (ENABLING), and Pillars fulfill the necessary conditions for reliability. The five aspects of environment-enabling conditions, infrastructure, demand, and sustainability have positive and significant relationships with the pillars of tourism destination competitiveness.

Table 6.	Path	Analysis	TTDI	2021
----------	------	----------	------	------

	Estimate (β)	Sample Mean (M)	SE	T-statistics	P values	Hypothesis
H1: CONN -> INF	0,947	0,950	0,014	67,703	0,000	Accepted
H2: INF -> Pillars	0,402	0,418	0,089	4,492	0,000	Accepted
H3: ENABLING -> Pillars	0,161	0,143	0,078	2,068	0,039	Accepted
H4: DEM -> Pillars	0,302	0,284	0,096	3,144	0,002	Accepted
H5: SUS -> Pillars	0,312	0,313	0,084	3,711	0,000	Accepted
H6a: Pillars -> Growth	-0,409	-0,441	0,149	2,749	0,006	Accepted
H6b: Pillars -> Value \$	0,600	0,626	0,094	6,370	0,000	Accepted
H6c: Pillars -> Share GDP	-0,357	-0,398	0,146	2,439	0,015	Accepted

Note. Fit Statistics for the structural models: p < 0.001; CFI = 0.606 (accepted if below the maximum limit of 0.9); NFI = 0.508 (accepted if below the maximum limit of 0.90); and RMSE = 0.065 (accepted if below the maximum limit of 0.08). Collinearity Statistics were examined using VIF values.

Source: own elaboration

T&T demand drivers, including available natural, cultural, and non-leisure resources, positively relate to tourism competitiveness in Asian countries (B=0,302). Similarly, enabling an environment that considers the business environment, safety and security, health and hygiene, human resources and labor, and ICT readiness also explains tourism competitiveness in Asia (B=0.161). This suggests that enabling an environment around tourist destinations is crucial for achieving attractive destinations. Moreover, infrastructure positively relates to tourism competitiveness, considering that air transport, ground and port infrastructure, and tourist service infrastructure allow for more competitive destinations. Infrastructure is supported by strong connectivity between Asia and its partners through appropriate logistics, air and ground transport, financial openness, and trade integration.





Moreover, T&T policy and enabling conditions suggest that government intervention in prioritization, international openness, and price competitiveness modestly influences competitiveness. Sustainability is crucial for competitive destinations, indicating the importance of environmental sustainability, socioeconomic resilience and conditions, and T&T demand pressures and impacts.

Moreover, the pillars (our measure of destination competitiveness) positively and significantly related to tourism performance (value \$), indicating that improving a destination's competitiveness is positively associated with tourism performance through more travelers, higher tourism receipts, and more job creation. The path analysis revealed significant effects on the relationship between tourism competitiveness (pillars) and tourism performance in terms of growth and economic contribution. However, the sign of this relationship was negative. One possible explanation is that highly competitive destinations might experience slower tourism growth rates, with tourism making a more negligible contribution to the overall economy than in less competitiveness may witness rapid growth in the tourism sector and an increasing contribution to the overall economy. This partially explains the race among emerging countries to promote tourism, expecting substantial impacts on job creation, spending, and investment. While the tourism sector in emerging Asian economies has experienced rapid growth, which has been closely linked to continuous improvements in tourist destinations (Anser et al., 2022; Esquivias et al., 2021; Heriqbaldi et al., 2023; Traskevich & Fontanari, 2023; Stankova et al. 2019), it has sometimes lagged behind those in more developed peer economies.

4.4. Model 2: Tourism Indicators – inner structural model

In Model 2, we incorporated several indicators (Figure 2). Like Model 1, we conducted all the necessary tests to validate the reliability, validity, and consistency of the data, constructs, and model (overall) (Table 7). Our research findings highlight the significance of the hypothesized aspects of destination competitiveness in shaping what we call "tourism pillars." These pillars include complementary conditions, core resources, destination management, infrastructure, connectivity, and sustainability, all of which contribute significantly to the development of competitive tourism destinations.

	Outer	Cronbach's	rho A	Composite	Average Variance
	Loadings	Alpha		Reliability	Extracted (AVE)
Complementary	Conditions	0.864	0.871	0.898	0.595
CC1	0,783				
CC2	0,760				
CC3	0,765				
CC5	0,847				
CC7	0,785				
CC8	0,680				
CC1	0,783				
Connectivity		0.709	0.800	0.831	0.624
CONNECT1	0,700				
FTAs	0,773				
Logistic	0,885				
Core Resources		0.844	0.978	0.887	0.663
CR1	0,820				
CR2	0,795				
CR3	0,767				

Table 7. Validity and reliability test results from tourism competitiveness indicators



JOURNAL OF TOURISM AND SERVICES Issue 28, volume 15, ISSN 1804-5650 (Online) www.jots.cz



CR5	0,871				
Destination ma	nagement	0.808	0.848	0.872	0.633
DM4	0,719				
DM6	0,797				
DM7	0,926				
Service	0,724				
Growt	h	0.788	0.859	0.863	0.615
BTS1	0,834				
DTS1	0,910				
GSTT1	0,671				
TTDCGDP1	0,696				
Infrastruc	cture	0.836	0.838	0.901	0.752
ICT	0,853				
INF1	0,881				
INF4	0,868				
Openne	ss	0.741	0.991	0.844	0.656
FDI_GDP	0,940				
Finance_Opennes	0,517				
Trade_GDP	0,904				
Pillars	8	0.774	0.818	0.857	0.605
ENV	0,804				
INF	0,958				
SUST	0,667				
TT_Demand	0,641				
Policy	7	0.768	0.867	0.856	0.669
Openness	0,643				
Policy	0,908				
Priority	0,877				
Share G	DP	0.715	0.845	0.830	0.622
GSTT2	0,891				
TTDCGDP2	0,710				
VE1	0,752				
Value	\$	0.813	0.906	0.879	0.709
TTCR5	0,891				
Rear	0,729				
Arrivals	0,895				

Source: own elaboration

Table 8 presents the results of the discriminant validity tests following Fornell and Larcker (1981). The presence of appropriate correlations among the latent constructs validated the discriminant validity of the measurement model, indicating the conceptual model's acceptability.

Table 8. Fornell-Larcker criteria for tourism competitiveness model employing T&T Competitiveness Report (TTCR 2021)

	CONN	DEM	ENABLING	ENV	Growth	INF	PILLARS TTDI	Pillars	SUS	Share GDP	Value \$
CONN	0.839										
DEM	0.614	0.854									
ENABLING	0.713	0.168	0.848								

Scopus[®]

JOURNAL OF TOURISM AND SERVICES Issue 28, volume 15, ISSN 1804-5650 (Online) www.jots.cz



ENV	0.720	0.180	0.702	0.847							
Growth	0.234	0.106	0.432	0.431	0.785						
INF	0.814	0.606	0.769	0.778	0.378	0.829					
PILLARS TTDI	0.822	0.676	0.782	0.790	0.358	0.803	0.819				
Pillars	0.811	0.376	0.813	0.816	0.448	0.807	0.802	0.872			
SUS	0.669	0.227	0.828	0.828	0.512	0.721	0.792	0.815	0.802		
Share GDP	0.256	0.389	0.234	0.239	0.323	0.334	0.314	0.262	0.206	0.830	
Value \$	0.554	0.757	0.255	0.262	0.065	0.514	0.571	0.352	0.157	0.136	0.886

Source: own elaboration

Core resources such as natural sites, cultural attractions, meeting venues, and unique species also significantly explain tourism competitiveness. This finding suggests that Asian countries can enhance competitiveness by improving and promoting these attractions, consistent with the findings of Heriqbaldi et al. (2023). Infrastructure, which includes airports, information and communication technology (ICT), and ground transportation facilities, is positively influenced by air connectivity, trade integration, and logistic performance. Infrastructure is positively related to tourism competitiveness, implying that countries can benefit by investing in more appropriate infrastructure, which is consistent with earlier findings for the UAE (Michael et al., 2019), top tourist destinations (Fernández et al., 2020), Indonesia (Santoso et al. 2023), and East Asia and the Pacific (Adeleye, 2023).

However, destination management exhibits a positive and significant relationship with the pillars of competitiveness. Tourism services, quality of education, training, and environmental management play significant roles in effective destination management. Moreover, destination management is enhanced by complementary conditions such as the availability of sufficient hotel rooms, ATMs, proper sanitation, Internet access, and mobile broadband, which positively and significantly relate to tourism competitiveness. Destination management and complementary conditions work synergistically to create attractive, competitive tourist destinations. Previous research has highlighted the significant influence of education, training, and skills on destination success (Fahlevi et al., 2020).

Moreover, we observed a positive relationship between improvements in tourism competitiveness and various tourism performance constructs (Table 9), including connectivity, tourism destination performance (international arrivals, domestic visitors, receipts, and job creation), and financial performance (business tourism spending, capital investment, contribution to GDP, and foreign spending). However, an intriguing finding emerged regarding the relationship between tourism pillars and performance growth and sharing. Surprisingly, we noticed a negative association, suggesting that countries with higher competitiveness indices might experience slower growth in tourism receipts and make fewer substantial contributions to economic factors such as exports and expenditure.

Reaching high levels of competitiveness may necessitate the exploration of other factors that foster growth in the tourism sector and its substantial contribution to the economy. Prior research has indicated that a destination's competitiveness, as measured by TTDI indicators, does not yield statistically significant effects on the relationship between tourism and economic growth (Webster & Ivanov, 2014). Similarly, top destinations based on tourism competitiveness do not necessarily reflect sizeable economic contributions (Croes & Kubickova, 2013; González-Rodríguez et al. 2023; Nazmfar et al., 2019). This observation aligns with previous studies that have highlighted how certain aspects of tourism, such as imports catering to tourists, repatriation of investment returns, and concentration of foreign expenditures in foreign-owned or large businesses (e.g., transportation and accommodation; Esquivias et al. (2021), may lead to adverse economic outcomes for destinations (González-Rodríguez et al. 2023; Tamene & Wondirad, 2019).



	Estimate (β)	Sample Mean (M)	SE	T- statistics	P Values	Hypothesis
H7: Connectivity -> Infrastructure	0.796	0.801	0.048	16.429	0.000	Accepted
H8: Complementary Conditions -> Destination management	0.842	0.853	0.033	25.344	0.000	Accepted
H9: Infrastructure -> Pillars	0.342	0.373	0.141	2.431	0.015	Accepted
H10: Destination management -> Pillars	0.493	0.449	0.144	3.437	0.001	Accepted
H11: Core Resources -> Pillars	0.329	0.310	0.095	3.476	0.001	Accepted
H12: Openness -> Share GDP	0.488	0.511	0.219	2.226	0.026	Accepted
H13: Share GDP -> Policy	0.442	0.487	0.170	2.600	0.010	
H14a: Pillars -> Growth	-0.406	-0.447	0.151	2.680	0.008	Accepted
H14b: Pillars -> Value \$	0.576	0.620	0.088	6.513	0.000	Accepted
H14c: Pillars -> Share GDP	-0.388	-0.346	0.156	2.494	0.013	Accepted

Table 9. Path Analysis TTCR

Notes: Fit Statistics for structural models: p < 0.001; CFI = 0.501 (accepted if below the maximum limit of 0.9); NFI = 0.458 (accepted if below the maximum limit of 0.90); RMSE = 0.06 (accepted if below the maximum limit of 0.08). Collinearity Statistics (VIF) verified (no issues detected).

Source: own elaboration

Another noteworthy finding was the negative relationship between the construct of openness, encompassing factors like Foreign Direct Investment (FDI) inflows, financial openness, and the share of trade to GDP, with both tourism's financial and tourism performance (although this was not statistically significant). This suggests that higher levels of openness might be associated with lower financial performance indicators such as foreign spending, capital investments, and tourism contributions to GDP. One possible explanation for this observation is that countries experiencing substantial FDI inflows and increased financial openness may rely more heavily on non-tourism sectors such as manufacturing, finance, or primary industries. Consequently, focusing on these sectors could divert attention and resources from the tourism industry, potentially leading to reduced financial performance in the tourism sector. Another possibility is that an increase in FDI and exports may strengthen a country's currency, resulting in increased costs for foreign visitors interested in tourism products and services. This exchange rate impact can negatively influence tourism demand, and consequently impact financial performance in the tourism sector. Countries with high levels of FDI inflows and significant export orientations may prioritize the development of manufacturing and other sectors over the tourism industry. Consequently, less investment and promotion may occur in the tourism sector, leading to a detrimental effect on international tourism receipts.

5. Discussion

These results indicate significant factors that may contribute to more attractive tourism destinations. However, as countries compete to attract visitors, they must design appropriate strategies to maximize the advantages of their destinations. It is common for countries to utilize policies that facilitate more efficient tourist destination management, promote the attributes of these destinations through targeted marketing campaigns, invest in tourism infrastructure projects that enhance access to popular locations, sponsor a variety of events (such as cultural, sporting, and business), offer visa exemptions, improve various facilities and services for tourists (such as Internet access, healthcare, and business environments), focus on connectivity projects both within and outside of national borders, and undertake a range of other initiatives. Despite these efforts, our study found that not all of these





initiatives necessarily resulted in an increase in tourist arrivals, jobs, revenue, or acceleration of growth in the tourism sector.

Empirical evidence indicates how some countries have developed successful strategies to capture market shares. Brazil branded its attractions as a Dutch holiday destination (Benedetti et al., 2011), Iran portrays itself as a more open country (Nematpour et al., 2022), the UK rebranded its destination during the London Olympics (Bourgeois, 2019), Portugal developed a strong identity as a tourism destination with rich heritage (Nobre & Sousa, 2022), Korea and Spain have labeled their tourism sector as a smart destination (Koo et al., 2016; Rucci et al., 2022) by embracing IT, and natural, rural, and cultural tourism in Africa (Mangachena et al., 2022), among other cases. Key resources must be effectively linked with supporting factors to achieve optimal performance. Several factors, excluding the location's resources, are likely to affect a country's long-term competitiveness (Song et al., 2023). Governments can enhance their competitiveness and achieve prosperity by leveraging both economic and noneconomic factors. Azzopardi and Nash (2016) noted that more than 60 factors have significant power to explain tourism destination competitiveness in Malta.

Both tourism models show that the constructs measuring complementary conditions and sustainability in tourism have a significant positive correlation with connectivity and overall tourism performance. This implies that better complementary conditions and sustainability can improve tourism performance and increase connectivity. This finding is consistent with earlier studies noting the significant effect of complementary aspects on shaping tourism competitiveness in cases like Iran (Nematpour et al., 2022), Indonesia (Lesmana et al., 2022), Malta (Azzopardi & Nash, 2016), and the UAE (Michael et al., 2019), among others (Hanafiah & Zulkifly, 2019). Similarly, sustainability can positively impact tourism destinations, consistent with previous evidence (Michael et al., 2019; Rodríguez-Díaz & Pulido-Fernández, 2019). However, complementary conditions and sustainability do not significantly affect economic performance (share and growth aspects of destinations). This finding holds significant importance as it suggests that enhancing tourism competitiveness in Asia may not necessarily improve residents' well-being. This aspect is vital when evaluating the overall impact of tourism on the well-being of local communities (Dwyer, 2022a). Gómez-Vega & J Picazo-Tadeo (2019) also observed that competitive destinations often result in higher value added per visitor (as reflected in the value indicator). However, this may not necessarily translate into increased growth or greater economic contribution to the destination.

In both models, tourism competitiveness indices can help countries enhance their competitiveness and attract tourists, consistent with earlier studies (González-Rodríguez et al. 2023; Nazmfar et al. 2019; Özer, M et al., 2022). However, their impact may exhibit a decreasing trend over time. This insight underscores the importance of highly competitive countries exploring additional avenues to promote further growth in the tourism sector and maximize its positive impact on the broader economy, including employment and exports.

Our findings are consistent with those of previous studies highlighting infrastructure's positive impact on tourism competitiveness, suggesting its ongoing relevance to flourishing destinations (Azzopardi & Nash, 2016; Domínguez Vila et al., 2015; Lesmana et al., 2022; Michael et al., 2019; Sul et al., 2020; Adeleye, 2023; Zainuddin et al., 2012). Similarly, core resources are crucial in a tourist destination's competitiveness, often manifested in attractive natural resources (Mangachena et al., 2022), cultural aspects (Heriqbaldi et al., 2023; Nobre & Sousa, 2022), non-natural leisure, and events (Moradi et al., 2022; Bourgeois, 2019), among others (Michael et al., 2019). Resources (natural and cultural) have been identified as the indicators with the largest contribution to tourism competitiveness (Gómez-Vega & J Picazo-Tadeo, 2019).

However, a negative association exists between core resources, infrastructure, and tourism growth. This could be attributed to the fact that countries with less advanced infrastructure experience higher tourism growth rates than those with more developed infrastructure. This trend highlights the growing interest of developing countries in utilizing tourism as a catalyst for economic growth (Ibrahim





et al. 2022; Özer et al., 2022). While initial efforts to enhance infrastructure may substantially affect tourism arrivals, the marginal effect of infrastructure improvements may decrease over the long term.

By contrast, destination management significantly and positively influences tourism performance and connectivity, underscoring its importance in enhancing overall tourism outcomes. This highlights the need for effective coordination between tourism authorities and connectivity at national and international levels to have a meaningful impact on destination performance, as noted in earlier studies (Gómez-Vega & J Picazo-Tadeo, 2019). Moreover, several studies conducted in various regions, such as Korea (Koo et al., 2016; Sul et al., 2020), Iran (Nematpour et al., 2022), Malta (Azzopardi & Nash, 2016), Indonesia (Ginting et al., 2023), and others (Drakulić Kovačević et al., 2018), have emphasized the favorable effects of destination management. This suggests that government interventions aimed at improving support services, providing hospitality-related training, and enhancing a destination's image can significantly boost the attractiveness of a destination for potential tourists.

A key factor contributing positively to tourism competitiveness is the availability of an enabling environment for tourism, which is characterized by favorable ICT, business environments, health facilities, and labor market conditions. This study offers new insights into the function of enabling factors in competitiveness, a finding that has been inconclusive in several studies (Michael et al., 2019). Nevertheless, the effect remained relatively modest, which is consistent with findings from prior research. Enabling aspects have been successful drivers of tourism in several regions, and can be further enhanced in Asia. Improving ICT has been shown to benefit various aspects of the tourism sector. For example, it can enhance tourism services (Le & Tran, 2023), improve leisure experiences (Herrera-Prado et al., 2023), provide easier access to information (Anser et al., 2022; Rucci et al., 2022), increase tourism resilience (Traskevich & Fontanari, 2023), increase sectoral competitiveness (Sul et al., 2020), drive innovation (Santoso et al. 2023), provide sectoral growth (Adeleye, 2023), and lead to more effective marketing campaigns (Ide, 2021). Previous studies have highlighted the significance of health in driving tourism receipts (Konstantakopoulou, 2022), indicating that Asian economies can attract more visitors by enhancing the accessibility and quality of health services.

5. Conclusions

We employed two models to examine the relationship between destination factors and tourism competitiveness. Similarly, we analyzed the potential link between tourism competitiveness (TDC) measures and tourism performance using indicators of destination performance, economic impact, and growth. Our analysis was based on data from the TTDI and TDI indicators for 35 Asian countries. Although numerous studies have devised models, indicators, and measures to evaluate the competitiveness of tourist destinations, relatively few have investigated whether improving competitiveness leads to better economic, financial, and tourism outcomes.

Overall, the competitive aspects of tourism have varying effects on tourism performance indicators. Complementary conditions and sustainability positively influence connectivity and overall tourism performance but may not be the primary drivers of tourism or economic growth. Core resources and infrastructure have vital impacts on connectivity, performance share, tourism destination performance, and financial performance, although they may also obstruct tourism growth. Conversely, destination management is essential in enhancing overall tourism performance and connectivity. The findings underscore the crucial role of core destination resources and infrastructure in achieving high tourism performance, particularly in driving key outcomes such as increased tourist numbers, higher tourism revenue, and the creation of jobs. These elements contribute substantially to the tourism sector's financial impact and overall contribution to the economy, encompassing expenditures, exports, business activities, and GDP growth.





These findings highlight significant practical implications for nations seeking to enhance their tourism sectors. First, countries with appealing and diverse resources, whether cultural, natural, or specialized, hold an advantage in fostering a competitive tourism environment. Governments and tourism authorities can leverage these inherent assets to create unique selling points, drawing on diverse travel segments and experiences. Second, countries can separate themselves from the global tourism market by focusing on enhancing and preserving these assets, creating a distinctive space to attract visitors seeking authentic, memorable experiences. Third, strategic investments in the development of core resources, infrastructure, connectivity, tourism services, and support is yield direct economic benefits. Such investments generate a ripple effect, drawing more tourists, driving higher revenue, and creating job opportunities. Additionally, it stimulates ancillary business activities, such as hospitality, local crafts, and services, thus contributing significantly to local and national economies. Fourth, preserving and highlighting appealing resources not only elevates the destination's competitiveness, but also emphasizes the importance of responsible tourism. Balancing growth with conservation ensures long-term positive impacts on the environment and local communities. Fifth, governments can explore partnerships with private entities, local communities, and international organizations to facilitate the development, maintenance, and preservation of these resources. Sixth, although initial investment may require substantial resources, the long-term economic benefits of a thriving tourism sector are significant. Over time, higher revenues, increased employment opportunities, and a robust local economy could outweigh the initial investments.

Some limitations and possible future directions in the current study pertain to the sample, methods, and identified factors. We used a sample of 35 Asian countries for our model; however, it is recommended to run a similar model using a different sample to compare the results and extrapolate possible public policies. We employed a PLS-SEM method to uncover the dynamic relations amongst factors and indicators of tourism performance, which is generally considered ideal; however, refinements of the method, such as PLS-PM, using higher-order constructs (Crocetta et al., 2021), discriminant statistical analysis (Gabor et al., 2021), or the dual-stage analysis of SEM and Artificial Neural Networks (ANN) in order to provide linear and non-compensatory relationships among constructs (Albahri et al., 2022) could lead to more adjusted results. A final limitation and open line of work is the identification of other factors that could contribute to tourism performance and city development.

Tourism competitiveness indicators positively impact tourist arrivals, revenue, and jobs. However, their contributions to economic growth and structural factors, such as the share of GDP, expenditure growth, investment, and exports, require further research. Countries must develop comprehensive destination strategies that effectively integrate their resources and capabilities while leveraging key services to enhance their competitive advantages. To create thriving destinations, it is essential to focus on developing destination images and branding, specialize in tourism-related services, and support the tourism sector through human, physical, and intangible assets. Although tourism competitiveness indicators offer valuable insights, they may not sufficiently guide tourism growth. This highlights the importance of tailoring strategies to each destination's specific needs and strengths to foster sustainable and robust tourism growth. By adopting such tailored approaches, countries can exploit the full potential of their tourism sectors and create flourishing destinations. Policymakers should closely consider tourism competitiveness indicators like TTDI and TTCR when designing tourism policies. However, our findings do not show that improvements in these indicators necessarily result in a larger share of tourism in GDP or faster sectoral growth.

References

JOURNAL OF TOURISM AND SERVICES Issue 28, volume 15, ISSN 1804-5650 (Online) www.jots.cz

Scopus



- 1. Adeleye, B. N. (2023). Re-examining the tourism-led growth nexus and the role of information and communication technology in East Asia and the Pacific. *Heliyon*, 9(2). https://doi.org/10.1016/j.heliyon.2023.e13505
- Aguiar-Barbosa, A. de P., Chim-Miki, A. F., & Kozak, M. (2021). Two decades of evolution in tourism competitiveness: A co-word analysis. *International Journal of Tourism Cities*, 7(2), 435–462. <u>https://doi.org/10.1108/IJTC-10-2020-0224</u>
- Albahri, A. S., Alnoor, A., Zaidan, A. A., Albahri, O. S., Hameed, H., Zaidan, B. B., ... & Yass, A. A. (2022). Hybrid artificial neural network and structural equation modelling techniques: a survey. *Complex & Intelligent Systems*, 8(2), 1781-1801. https://doi.org/10.1007/s40747-021-00503
- Alkathiri, N. A., & Soliman, M. (2020). Examining foreign direct investment determinants of tourism industry in Oman and Egypt: The moderating role of investment environment. *International Journal of Finance & Economics*, ijfe.2396. <u>https://doi.org/10.1002/ijfe.2396</u>
- Anser, M. K., Adeleye, B. N., Tabash, M. I., & Tiwari, A. K. (2022). Services trade–ICT– tourism nexus in selected Asian countries: New evidence from panel data techniques. *Current Issues in Tourism*, 25(15), 2388–2403. <u>https://doi.org/10.1080/13683500.2021.1965554</u>
- 6. Azzopardi, E., & Nash, R. (2016). A framework for island destination competitiveness perspectives from the island of Malta. *Current Issues in Tourism*, 19(3), 253–281. https://doi.org/10.1080/13683500.2015.1025723
- Bamgbade, J. A., Kamaruddeen, A. M., Nawi, M. N. M., Yusoff, R. Z., & Bin, R. A. (2018). Does government support matter? Influence of organizational culture on sustainable construction among Malaysian contractors. *International Journal of Construction Management*, 18(2), 93–107. <u>https://doi.org/10.1080/15623599.2016.1277057</u>
- 8. Barclay, D., Higgins, C., & Thompson, R. (1995). The partial least squares (PLS) approach to casual modeling: Personal computer adoption ans use as an Illustration.
- Benedetti, J., Çakmak, E., & Dinnie, K. (2011). The competitive identity of Brazil as a Dutch holiday destination. *Place Branding and Public Diplomacy*, 7(2), 107–115. https://doi.org/10.1057/pb.2011.10
- 10. Bourgeois, T. (2019). London 2012 Olympics: exercises in cultural diplomacy. *Place Branding and Public Diplomacy*, 15(3), 198-205. https://doi.org/10.1057/s41254-017-0084-4
- 11. Cenfetelli, R. T., & Bassellier, G. (2009). Interpretation of formative measurement in information systems research. *MIS Quarterly*, 689–707.
- Chin, W. W., Marcolin, B. L., & Newsted, P. R. (2003). A Partial Least Squares Latent Variable Modeling Approach for Measuring Interaction Effects: Results from a Monte Carlo Simulation Study and an Electronic-Mail Emotion/Adoption Study. *Information Systems Research*, 14(2), 189– 217. <u>https://doi.org/10.1287/isre.14.2.189.16018</u>
- 13. Civelek, M., Krajčík, V., & Fialova, V. (2023a). The impacts of innovative and competitive abilities of SMEs on their different financial risk concerns: System approach. *Oeconomia Copernicana*, 14(1), 327–354. https://doi.org/10.24136/oc.2023.009
- 14. Civelek, M., Krajčík, V., & Ključnikov, A. (2023b). The impacts of dynamic capabilities on SMEs' digital transformation process: The resource-based view perspective. *Oeconomia Copernicana*. https://doi.org/10.24136/oc.2023.019
- 15. Croes, R. (2011). Measuring and explaining competitiveness in the context of small island destinations. *Journal of Travel Research*, 50(4), 431–442.
- Croes, R., & Kubickova, M. (2013). From potential to ability to compete: Towards a performance-based tourism competitiveness index. *Journal of Destination Marketing & Management*, 2(3), 146–154. <u>https://doi.org/10.1016/j.jdmm.2013.07.002</u>
- 17. Croes, R., & Rivera, M. A. (2010). Testing the empirical link between tourism and competitiveness: Evidence from Puerto Rico. *Tourism Economics*, 16(1), 217–234.

JOURNAL OF TOURISM AND SERVICES Issue 28, volume 15, ISSN 1804-5650 (Online) www.jots.cz

Scopus



- Crocetta, C., Antonucci, L., Cataldo, R. et al. Higher-Order PLS-PM Approach for Different Types of Constructs. Soc Indic Res 154, 725–754 (2021). <u>https://doi.org/10.1007/s11205-020-02563-w</u>
- 19. Crouch, G. I. (2011). Destination competitiveness: An analysis of determinant attributes. *Journal of Travel Research*, 50(1), 27–45.
- 20. Crouch, G. I., & Ritchie, J. R. B. (1999). Tourism, Competitiveness, and Societal Prosperity. Journal of Business Research, 44(3), 137–152. <u>https://doi.org/10.1016/S0148-2963(97)00196-3</u>
- Domínguez Vila, T., Darcy, S., & Alén González, E. (2015). Competing for the disability tourism market A comparative exploration of the factors of accessible tourism competitiveness in Spain and Australia. *Tourism Management*, 47, 261–272. https://doi.org/10.1016/j.tourman.2014.10.008
- 22. Drakulić Kovačević, N., Kovačević, L., Stankov, U., Dragićević, V., & Miletić, A. (2018). Applying destination competitiveness model to strategic tourism development of small destinations: The case of South Banat district. *Journal of Destination Marketing & Management*, 8, 114–124. <u>https://doi.org/10.1016/j.jdmm.2017.01.002</u>
- 23. Dwyer, L. (2022a). Destination competitiveness and resident well-being. *Tourism Management Perspectives*, 43, 100996. <u>https://doi.org/10.1016/j.tmp.2022.100996</u>
- 24. Dwyer, L. (2022b). Productivity, Destination Performance, and Stakeholder Well-Being. *Tourism and Hospitality*, *3*(3), 618–633. <u>https://doi.org/10.3390/tourhosp3030038</u>
- 25. Dwyer, L., & Kim, C. (2003). Destination Competitiveness: Determinants and Indicators. *Current Issues in Tourism*, 6(5), 369–414. <u>https://doi.org/10.1080/13683500308667962</u>
- 26. Esquivias, M. A., Sugiharti, L., Rohmawati, H., & Sethi, N. (2021). Impacts and implications of a pandemic on tourism demand in Indonesia. *Economics & Sociology*, 14(4), 133–150. https://doi.org/10.14254/2071-789X.2021/14-4/8
- 27. Fahlevi, M., Rabiah, A. S., Pradipta, I. A., Marta, A., & Dipo, F. (2020). Tourism and absorption of the labor force in Indonesia: a strategy for development. In E3S Web of Conferences (Vol. 202, p. 16001). EDP Sciences. 10.1051/e3sconf/202020216001
- Fernández, J. A. S., Azevedo, P. S., Martín, J. M. M., & Martín, J. A. R. (2020). Determinants of tourism destination competitiveness in the countries most visited by international tourists: Proposal of a synthetic index. Tourism Management Perspectives, 33, 100582. <u>https://doi.org/10.1016/j.tmp.2019.100582</u>
- 29. Fornell, C., & Larcker, D. F. (1981). Evaluating structural equation models with unobservable variables and measurement error. *Journal of Marketing Research*, 18(1), 39–50.
- 30. Gabor, M. R., Kardos, M., Cristache, N., Nastase, M., & Petrariu, I. R. (2021). Dynamic analysis of tourism competitiveness of the European countries based on discriminant statistical analysis.. Economic Computation & Economic Cybernetics Studies & Research, 55(3).
- Gavurova, B., Privara, A., Janikova, J., & Kovac, V. (2021). Quantification of Tourism Sector Parameters Related to Competitiveness of Countries According to Macroeconomic Indicators. *Journal of Competitiveness*, 13(3), 56–72. https://doi.org/10.7441/joc.2021.03.04
- Ginting, N. C. A., Sianturi, M., & Ariesa, Y. (2023). The influence of service quality and price on guest loyalty during the COVID-19 pandemic with guest satisfaction as an intervening variable at Hotel Grand Cityhall Medan. *Priviet Social Sciences Journal*, 3(7), 1– 8. <u>https://doi.org/10.55942/pssj.v3i7.227</u>
- Gómez-Vega, M., & J Picazo-Tadeo, A. (2019). Ranking world tourist destinations with a composite indicator of competitiveness: To weigh or not to weigh? *Tourism Management*, 72, 281–291. <u>https://doi.org/10.1016/j.tourman.2018.11.006</u>
- 34. González-Rodríguez, M. R., Díaz-Fernández, M. C., & Pulido-Pavón, N. (2023). Tourist destination competitiveness: An international approach through the travel and tourism





competitiveness index. Tourism Management Perspectives, 47, 101127. https://doi.org/10.1016/j.tmp.2023.101127

- 35. Gooroochurn, N., & Sugiyarto, G. (2005). Competitiveness Indicators in the Travel and Tourism Industry. *Tourism Economics*, 11(1), 25–43. <u>https://doi.org/10.5367/0000000053297130</u>
- 36. Grissemann, U. S., & Stokburger-Sauer, N. E. (2012). Customer co-creation of travel services: The role of company support and customer satisfaction with the co-creation performance. *Tourism Management*, 33(6), 1483–1492.
- Hair, J. F., Sarstedt, M., Ringle, C. M., & Mena, J. A. (2012). An assessment of the use of partial least squares structural equation modeling in marketing research. *Journal of the Academy of Marketing Science*, 40(3), 414–433. <u>https://doi.org/10.1007/s11747-011-0261-6</u>
- 38. Hair Jr, J. F., Hult, G. T. M., Ringle, C. M., Sarstedt, M., Danks, N. P., & Ray, S. (2021). Partial least squares structural equation modeling (PLS-SEM) using R: A workbook. Springer Nature.
- 39. Hanafiah, M. H., & Zulkifly, M. I. (2019). Tourism destination competitiveness and tourism performance: A secondary data approach. *Competitiveness Review: An International Business Journal*, 29(5), 592–621. <u>https://doi.org/10.1108/CR-07-2018-0045</u>
- 40. Heath, E. T. (2003). Towards a model to enhance destination competitiveness: A Southern African perspective.
- Henseler, J., Hubona, G., & Ray, P. A. (2016). Using PLS path modeling in new technology research: Updated guidelines. *Industrial Management & Data Systems*, 116(1), 2–20. <u>https://doi.org/10.1108/IMDS-09-2015-0382</u>
- 42. Henseler, J., Ringle, C. M., & Sinkovics, R. R. (2009). The use of partial least squares path modeling in international marketing. In R. R. Sinkovics & P. N. Ghauri (Eds.), *Advances in International Marketing* (Vol. 20, pp. 277–319). Emerald Group Publishing Limited. https://doi.org/10.1108/S1474-7979(2009)0000020014
- 43. Heriqbaldi, U., Esquivias, M. A., & Agusti, K. S. (2023). The role of cultural distance in boosting international tourism arrivals in ASEAN: A gravity model. *Consumer Behavior in Tourism and Hospitality*, 18(1), 97–109. <u>https://doi.org/10.1108/CBTH-12-2021-0288</u>
- Herrera-Prado, A. L., Olmos-Martínez, E., & León-Castro, E. (2023). Impact of Information and Communication Technology on Tourist Spending. In M. Sahni, J. M. Merigó, W. Hussain, E. León-Castro, R. K. Verma, & R. Sahni (Eds.), *Mathematical Modeling, Computational Intelligence Techniques and Renewable Energy* (Vol. 1440, pp. 329–347). Springer Nature Singapore. <u>https://doi.org/10.1007/978-981-19-9906-2_27</u>
- Ibrahim, K. H., Handoyo, R. D., Wasiaturrahma, W., & Sarmidi, T. (2022). Services trade and infrastructure development: Evidence from African countries. *Cogent Economics & Finance*, 10(1), 2143147. <u>https://doi.org/10.1080/23322039.2022.2143147</u>
- 46. Ide, A. (2021). Tourism and ICT Solutions in the COVID-19 Era: A Comparison Between Japan and Sweden. *The Review of Socionetwork Strategies*, 15(1), 195–211. https://doi.org/10.1007/s12626-021-00072-x
- 47. International Monetary Fund Annual Report 2021: Build Forward Better. (2021). International Monetary Fund. <u>https://www.imf.org/external/pubs/ft/ar/2021/eng/</u>
- 48. Kim, N., & Wicks, B. E. (2010). Rethinking tourism cluster development models for global competitiveness.
- 49. Kim, Y. R., Liu, A., & Williams, A. M. (2022). Competitiveness in the visitor economy: A systematic literature review. *Tourism Economics*, 28(3), 817–842. https://doi.org/10.1177/13548166211034437
- 50. Konstantakopoulou, I. (2022). Does health quality affect tourism? Evidence from system GMM estimates. *Economic Analysis and Policy*, 73, 425–440. <u>https://doi.org/10.1016/j.eap.2021.12.007</u>
- Koo, C., Shin, S., Gretzel, U., Hunter, W. C., & Chung, N. (2016). Conceptualization of Smart Tourism Destination Competitiveness. *Asia Pacific Journal of Information Systems*, 26(4), 561–576. <u>https://doi.org/10.14329/apjis.2016.26.4.561</u>





- 52. Krivokapic, R. (2023). Perception of the Influence of the Essential Factors on the Competitiveness of Small Adriatic Seaports. *Transformations in Business & Economics*, 22(2),110-123.
- 53. Kunst, I., & Ivandić, N. (2021). The viability of the travel and tourism competitiveness index as a reliable measure of destination competitiveness: The case of the Mediterranean region. *European Journal of Tourism Research*, 27, 2704. <u>https://doi.org/10.54055/ejtr.v27i.2124</u>
- 54. Kurar, İ. (2021). Research on the determination of recreational experience preferences, expectations, and satisfaction levels of local people. *International Journal of Entrepreneurial Knowledge*, 9(1), 41-66. <u>https://doi.org/10.37335/ijek.v9i1.122</u>
- 55. Le, D. V., & Tran, T. Q. (2023). Heterogenous effects of ICT application on local tourism development: A quantile regression approach. *Journal of Policy Research in Tourism, Leisure and Events*, 1–23. <u>https://doi.org/10.1080/19407963.2023.2196727</u>
- Lesmana, H., Sugiarto, S., Yosevina, C., & Widjojo, H. (2022). A Competitive Advantage Model for Indonesia's Sustainable Tourism Destinations from Supply and Demand Side Perspectives. *Sustainability*, 14(24), 16398. <u>https://doi.org/10.3390/su142416398</u>
- 57. Mwinuka, L., & Mwangoka, V. C. (2023). Manufacturing sector's growth in Tanzania: Empirical lessons from macroeconomic factors, 1970–2021. *Cogent Economics & Finance*, 11(1), 2223419. https://doi.org/10.1080/23322039.2023.2223419
- 58. Leung, X. Y., & Baloglu, S. (2013). Tourism competitiveness of Asia Pacific destinations. *Tourism Analysis*, 18(4), 371–384.
- 59. Lohmöller, J.-B. (2013). Latent variable path modeling with partial least squares. Springer Science & Business Media.
- 60. Luštický, M., & Štumpf, P. (2021). Leverage points of tourism destination competitiveness dynamics. *Tourism Management Perspectives*, *38*, 100792. https://doi.org/10.1016/j.tmp.2021.100792
- 61. Mangachena, J. R., Chauvenet, A. L. M., & Pickering, C. M. (2022). Destination Images of Africa Reflected in Social Media Photographs. *Society & Natural Resources*, 35(11), 1170–1188. https://doi.org/10.1080/08941920.2022.2100023
- 62. Michael, N., Reisinger, Y., & Hayes, J. P. (2019). The UAE's tourism competitiveness: A business perspective. *Tourism Management Perspectives*, 30, 53-64. https://doi.org/10.1016/j.tmp.2019.02.002
- 63. Mior Shariffuddin, N. S., Azinuddin, M., Hanafiah, M. H., & Wan Mohd Zain, W. M. A. (2022). A comprehensive review on tourism destination competitiveness (TDC) literature. *Competitiveness Review: An International Business Journal*. <u>https://doi.org/10.1108/CR-04-2021-0054</u>
- 64. Moradi, E., Ehsani, M., Saffari, M., & Norouzi Seyed Hosseini, R. (2022). Developing an integrated model for the competitiveness of sports tourism destinations. *Journal of Destination Marketing & Management*, 26, 100743. <u>https://doi.org/10.1016/j.jdmm.2022.100743</u>
- Nadalipour, Z., Imani Khoshkhoo, M. H., & Eftekhari, A. R. (2019). An integrated model of destination sustainable competitiveness. *Competitiveness Review: An International Business Journal*, 29(4), 314–335. <u>https://doi.org/10.1108/CR-12-2017-0086</u>
- 66. Nazmfar, H., Eshghei, A., Alavi, S., & Pourmoradian, S. (2019). Analysis of travel and tourism competitiveness index in middle-east countries. *Asia Pacific Journal of Tourism Research*, 24(6), 501-513. <u>https://doi.org/10.1080/10941665.2019.1590428</u>
- Nematpour, M., Khodadadi, M., Makian, S., & Ghaffari, M. (2022). Developing a Competitive and Sustainable Model for the Future of a Destination: Iran's Tourism Competitiveness. *International Journal of Hospitality & Tourism Administration*, 1–33. <u>https://doi.org/10.1080/15256480.2022.2081279</u>

JOURNAL OF TOURISM AND SERVICES Issue 28, volume 15, ISSN 1804-5650 (Online) www.jots.cz





- 68. Nobre, H., & Sousa, A. (2022). Cultural heritage and nation branding multi stakeholder perspectives from Portugal. *Journal of Tourism and Cultural Change*, 20(5), 699–717. https://doi.org/10.1080/14766825.2021.2025383
- 69. Novais, M. A., Ruhanen, L., & Arcodia, C. (2018). Destination competitiveness: A phenomenographic study. Tourism management, 64, 324-334. https://doi.org/10.1016/j.tourman.2017.08.014
- 70. Özer, M., Küçüksakarya, S., & Maiti, M. (2022). Nexus of tourism demand, economic growth, and external competitiveness in leading tourist destination countries. *Tourism Management Perspectives*, 42, 100965. <u>https://doi.org/10.1016/j.tmp.2022.100965</u>
- Perez Leon, V. E., Perez, F., Contreras Rubio, I., & Guerrero, F. M. (2021). An approach to the travel and tourism competitiveness index in the Caribbean region. *International Journal of Tourism Research*, 23(3), 346-362. <u>https://doi.org/10.1002/jtr.2411</u>
- 72. Pulido-Fernández, J. I., & Rodríguez-Díaz, B. (2016). Reinterpreting the World Economic Forum's global tourism competitiveness index. *Tourism Management Perspectives*, 20, 131–140. https://doi.org/10.1016/j.tmp.2016.08.001
- 73. Ragab, A. M., & Meis, S. (2016). Developing environmental performance measures for tourism using a Tourism Satellite Accounts approach: A pilot study of the accommodation industry in Egypt. *Journal of Sustainable Tourism*, 24(7), 1007–1023. https://doi.org/10.1080/09669582.2015.1107078
- 74. Ritchie, J. B., & Crouch, G. I. (2003). The competitive destination: A sustainable tourism perspective. Cabi.
- 75. Rodríguez-Díaz, B., & Pulido-Fernández, J. I. (2019). Sustainability as a Key Factor in Tourism Competitiveness: A Global Analysis. Sustainability, 12(1), 51. https://doi.org/10.3390/su12010051
- 76. Rucci, A. C., Moreno-Izquierdo, L., Perles-Ribes, J. F., & Porto, N. (2022). Smart or partly smart? Accessibility and innovation policies to assess smartness and competitiveness of destinations. *Current Issues in Tourism*, 25(8), 1270-1288. https://doi.org/10.1080/13683500.2021.1914005.
- 77. Santoso S, Widyanty W, Nurhidajat R, Ramadhani Marfatah M, Mahmud G, Fahlevi M, Aljuaid M, Zhghenti T and Shahid D (2022) System dynamics modeling for developing an agrotourismcreative economy in the framework of the village innovation system. *Frontiers in Environmental Science*. 10:962235. doi: 10.3389/fenvs.2022.962235
- 78. Sarstedt, M., Ringle, C. M., Smith, D., Reams, R., & Hair, J. F. (2014). Partial least squares structural equation modeling (PLS-SEM): A useful tool for family business researchers. *Journal of Family Business Strategy*, 5(1), 105–115. <u>https://doi.org/10.1016/j.jfbs.2014.01.002</u>
- 79. Shariffuddin, N. S. M., Azinuddin, M., Yahya, N. E., & Hanafiah, M. H. (2023). Navigating the tourism digital landscape: The interrelationship of online travel sites' affordances, technology readiness, online purchase intentions, trust, and E-loyalty. *Heliyon*, 9(8). <u>https://doi.org/10.1016/j.heliyon.2023.e19135</u>
- 80. Song, H., Jiao, E., & Park, S. (2023). Sectoral productivity and destination competitiveness. *Annals of Tourism Research*, 103, 103645. https://doi.org/10.1016/j.annals.2023.103645
- 81. Stankova, M., Tsvetkov, T., & Ivanova, L. (2019). Tourist development between security and terrorism: empirical evidence from Europe and the United States. *Oeconomia Copernicana*, 10(2), 219-237. https://doi.org/10.24136/oc.2019.011
- Sul, H.-K., Chi, X., & Han, H. (2020). Measurement Development for Tourism Destination Business Environment and Competitive Advantages. *Sustainability*, 12(20), 8587. <u>https://doi.org/10.3390/su12208587</u>

JOURNAL OF TOURISM AND SERVICES Issue 28, volume 15, ISSN 1804-5650 (Online) www.jots.cz





- 83. Syofian, Markoni, Fauzan, & Lakoni, I. (2023). Investigating tourist visit behavior in Bengkulu, Indonesia: The role of satisfaction, trust, and price. *Journal of Economics and Business Letters*, 3(2), 18–21. https://doi.org/10.55942/jebl.v3i2.202
- Tamene, K., & Wondirad, A. (2019). Economic impacts of tourism on small-scale tourism enterprises (SSTEs) in Hawassa City, Southern Ethiopia. *International Journal of Tourism Sciences*, 19(1), 38–55. <u>https://doi.org/10.1080/15980634.2019.1592951</u>
- 85. Tenenhaus, M., Vinzi, V. E., Chatelin, Y.-M., & Lauro, C. (2005). PLS path modeling. Computational Statistics & Data Analysis, 48(1), 159–205. <u>https://doi.org/10.1016/j.csda.2004.03.005</u>
- Traskevich, A., & Fontanari, M. (2023). Tourism Potentials in Post-COVID19: The Concept of Destination Resilience for Advanced Sustainable Management in Tourism. *Tourism Planning & Development*, 20(1), 12–36. <u>https://doi.org/10.1080/21568316.2021.1894599</u>
- 87. UNDP. (2021). United Nations Development Programme. https://www.undp.org/publications/undp-annual-report-2021
- Webster, C., & Ivanov, S. (2014). Transforming competitiveness into economic benefits: Does tourism stimulate economic growth in more competitive destinations? *Tourism Management*, 40, 137–140. <u>https://doi.org/10.1016/j.tourman.2013.06.003</u>
- 89. Wei, Y., Liu, H., & Park, K. S. (2021). Examining the structural relationships among heritage proximity, perceived impacts, attitude and residents' support in intangible cultural heritage tourism. *Sustainability*, *13*(15), 8358. <u>https://doi.org/10.3390/su13158358</u>
- 90. Wong, K. K.-K. (2013). Partial least squares structural equation modeling (PLS-SEM) techniques using SmartPLS. *Marketing Bulletin*, 24(1), 1–32.
- 91. Zadeh Bazargani, R. H., & Kiliç, H. (2021). Tourism competitiveness and tourism sector performance: Empirical insights from new data. *Journal of Hospitality and Tourism Management*, 46, 73–82. <u>https://doi.org/10.1016/j.jhtm.2020.11.011</u>
- 92. Zainuddin, Z., Hilmy, P., Ghafar, A., Zahari, M. S. M., & Radzi, S. M. (2012). Destination competitiveness, tourism performance and resident's quality of life. Current Issues in Hospitality and Tourism: Research and Innovations, 497-501.

Brief description of Author/Authors:

Dr. Rudi Purwono

ORCID ID: https://orcid.org/0000-0002-4902-0745

Affiliation: Department of Economics, Airlangga University, Surabaya 60286, Indonesia,

Email: <u>rudipurwono@feb.unair.ac.id</u>

Full Professor of Economics at the Faculty of Economics and Business, Universitas Airlangga. He serves as Vice Dean at the Graduate School, Universitas Airlangga. He holds the position of Vice-Chair of the Surabaya Branch of the Indonesian Economists Association (ISEI). Indonesian Bureau of Economic Research (IBER). Former Member of the Economic Forum of the Ministry of Finance of the Republic of Indonesia and former Independent Commissioner of the Provincial Development Bank of East Java, Indonesia.

Dr. Miguel Angel Esquivias

ORCID ID: <u>https://orcid.org/0000-0002-1282-6163</u> Affiliation: Department of Economics, Airlangga University, Surabaya 60286, Indonesia, Email: <u>miguel@feb.unair.ac.id</u>





He serves as the Head of Research at the Economic Development Laboratory of the Faculty of Economics and Business at Airlangga University. Economic research consultancy services in the areas of Investment, Trade, and Industry to the National Ministry of Development in the Republic of Indonesia. Specialization in international economics focuses on trade, investment, and tourism. Former consultants at the Free Trade Agreement Center under the Ministry of Trade in the Republic of Indonesia. Members of the National System of Researchers of Mexico (CONACYT).

Dr. Lilik Sugiharti, corresponding author

ORCID ID: https://orcid.org/0000-0001-5156-7929

Affiliation: Department of Economics, Airlangga University, Surabaya 60286, Indonesia,

Email: sugiharti.lilik@feb.unair.ac.id

Associate Professor and previously served as the Head of the Department of Economics at Universitas Airlangga. Former Director of the Economic Development Laboratory at the Faculty of Economics and Business at Airlangga University. Academic expertise in the field of Economic Development, with a focus on critical areas such as poverty alleviation, labor economics, gender inclusivity, and enhancing competitiveness. Leading Researcher for several national competitive research grants from the Republic of Indonesia since 2012.

Ph.D Omar Rojas

ORCID ID: https://orcid.org/0000-0002-0681-3833

Affiliation: Facultad de Ciencias Económicas y Empresariales, Universidad Panamericana, Álvaro del Portillo 49, Zapopan 45010, Mexico;

Email: orojas@up.edu.mx

Professor in Decision Analysis and Business Analytics. Members of the National System of Researchers of Mexico (CONACYT). He serves as the Director of the Doctorate in Business Sciences at Universidad Panamericana, Mexico. Previously, he was a Research Director at the College of Business Sciences at Universidad Panamericana. Lead Data Scientist and co-founder of Intelica, a company specializing in scientific and software solutions.